UNIVERSITY of York

This is a repository copy of *Comorbidity* and health related quality of life in people with a chronic medical condition in randomised clinical trials:an individual participant data metaanalysis.

White Rose Research Online URL for this paper: <u>https://eprints.whiterose.ac.uk/id/eprint/194657/</u>

Version: Published Version

Article:

Butterly, Elaine, Hanlon, Peter, Shah, Anoop S. et al. (10 more authors) (2023) Comorbidity and health related quality of life in people with a chronic medical condition in randomised clinical trials:an individual participant data meta-analysis. Plos medicine. e1004154. ISSN: 1549-1277

https://doi.org/10.1371/journal.pmed.1004154

Reuse

This article is distributed under the terms of the Creative Commons Attribution (CC BY) licence. This licence allows you to distribute, remix, tweak, and build upon the work, even commercially, as long as you credit the authors for the original work. More information and the full terms of the licence here: https://creativecommons.org/licenses/

Takedown

If you consider content in White Rose Research Online to be in breach of UK law, please notify us by emailing eprints@whiterose.ac.uk including the URL of the record and the reason for the withdrawal request.



eprints@whiterose.ac.uk https://eprints.whiterose.ac.uk/



G OPEN ACCESS

Citation: Butterly EW, Hanlon P, Shah ASV, Hannigan LJ, McIntosh E, Lewsey J, et al. (2023) Comorbidity and health-related quality of life in people with a chronic medical condition in randomised clinical trials: An individual participant data meta-analysis. PLoS Med 20(1): e1004154. https://doi.org/10.1371/journal.pmed.1004154

Received: June 24, 2022

Accepted: December 9, 2022

Published: January 17, 2023

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pmed.1004154

Copyright: © 2023 Butterly et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data that support the findings of this study are available from Clinical Study Data Request and the Yale Open Data Access repositories but restrictions apply to the availability of these data, which were used under license for the current study, and so are not RESEARCH ARTICLE

Comorbidity and health-related quality of life in people with a chronic medical condition in randomised clinical trials: An individual participant data meta-analysis

Elaine W. Butterly[®]¹, Peter Hanlon[®]¹, Anoop S. V. Shah[®]², Laurie J. Hannigan[®]^{3,4,5}, Emma McIntosh¹, Jim Lewsey¹, Sarah H. Wild[®]⁶, Bruce Guthrie[®]⁶, Frances S. Mair¹, David M. Kent⁷, Sofia Dias[®]⁸, Nicky J. Welton⁴, David A. McAllister[®]¹*

 School of Health and Wellbeing, College of Medical, Veterinary and Life Sciences, University of Glasgow, Glasgow, United Kingdom, 2 London School of Hygiene and Tropical Medicine, London, United Kingdom,
Nic Waals Institute, Lovisenberg Diaconal Hospital, Oslo, Norway, 4 Population Health Sciences, Bristol Medical School, University of Bristol, Bristol, United Kingdom, 5 Department of Mental Disorders, Norwegian Institute of Public Health, Oslo, Norway, 6 Usher Institute, University of Edinburgh, Edinburgh, United Kingdom, 7 Predictive Analytics and Comparative Effectiveness Center, Tufts Medical Center/Tufts University School of Medicine, Boston, Massachusetts, United States of America, 8 Centre for Reviews and Dissemination, University of York, York, United Kingdom

* david.mcallister@glasgow.ac.uk

Abstract

Background

Health-related quality of life metrics evaluate treatments in ways that matter to patients, so are often included in randomised clinical trials (hereafter trials). Multimorbidity, where individuals have 2 or more conditions, is negatively associated with quality of life. However, whether multimorbidity predicts change over time or modifies treatment effects for quality of life is unknown. Therefore, clinicians and guideline developers are uncertain about the applicability of trial findings to people with multimorbidity. We examined whether comorbidity count (higher counts indicating greater multimorbidity) (i) is associated with quality of life at baseline; (ii) predicts change in quality of life over time; and/or (iii) modifies treatment effects on quality of life.

Methods and findings

Included trials were registered on the United States trials registry for selected index medical conditions and drug classes, phase 2/3, 3 or 4, had \geq 300 participants, a nonrestrictive upper age limit, and were available on 1 of 2 trial repositories on 21 November 2016 and 18 May 2018, respectively. Of 124 meeting these criteria, 56 trials (33,421 participants, 16 index conditions, and 23 drug classes) collected a generic quality of life outcome measure (35 EuroQol-5 dimension (EQ-5D), 31 36-item short form survey (SF-36) with 10 collecting both). Blinding and completeness of follow up were examined for each trial.

Using trials where individual participant data (IPD) was available from 2 repositories, a comorbidity count was calculated from medical history and/or prescriptions data. Linear

publicly available. Data are however available by directly applying to these repositories via application process to their respective independent data access committees. All data released from the respective safe havens (Clinical Study Data Request and the Yale Open Data Access) has been made available at https://github.com/ ChronicDiseaseEpi/como_qol_public.

Funding: DM is funded via an Intermediate Clinical Fellowship and Beit Fellowship from the Wellcome Trust, who also supported other costs related to this project such as data access costs and database licenses ("Treatment effectiveness in multimorbidity: Combining efficacy estimates from clinical trials with the natural history obtained from large routine healthcare databases to determine net overall treatment Benefits." - 201492/Z/16/Z. SD is supported by the Medical Research Council [grant no. MR/R025223/1]. LH is supported by the South-Eastern Norway Regional Health Authority (#2020023, #2019097). PH is funded through a Clinical Research Training Fellowship from the Medical Research Council (Grant reference: MR/ S021949/1). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests. NJW has received honoraria for training and masterclasses from: Association of British Pharmaceutical Industries, Campbell Ireland, Centre for Global Development, NICE International and NICE Scientific Advice, All Wales Therapeutics and Toxicology Centre, University of Leuven. NJW has delivered training for Takeda, ICON plc, and University of Galway for which payment was made to her institution. DM has received funding for this work from the Wellcome Trust.

Abbreviations: ATC, anatomic therapeutic chemical; CSDR, Clinical Study Data Request; EQ-5D, EuroQol-5 dimension; IPD, individual participant data; MCS, mental component score; NCT, national clinical trial; PCS, physical component score; SD, standard deviation; SLE, systemic lupus erythematosus; TNF-α, tumour necrosis factor α; YODA, Yale University Open Data Access. regressions were fitted for the association between comorbidity count and (i) quality of life at baseline; (ii) change in quality of life during trial follow up; and (iii) treatment effects on quality of life. These results were then combined in Bayesian linear models. Posterior samples were summarised via the mean, 2.5th and 97.5th percentiles as credible intervals (95% CI) and via the proportion with values less than 0 as the probability (P_{Bayes}) of a negative association.

All results are in standardised units (obtained by dividing the EQ-5D/SF-36 estimates by published population standard deviations). Per additional comorbidity, adjusting for age and sex, across all index conditions and treatment comparisons, comorbidity count was associated with lower quality of life at baseline and with a decline in quality of life over time (EQ-5D -0.02 [95% CI -0.03 to -0.01], P_{Bayes} > 0.999). Associations were similar, but with wider 95% CIs crossing the null for SF-36-PCS and SF-36-MCS (-0.05 [-0.10 to 0.01], P_{Bayes} = 0.956 and -0.05 [-0.10 to 0.01], P_{Bayes} = 0.966, respectively). Importantly, there was no evidence of any interaction between comorbidity count and treatment efficacy for either EQ-5D or SF-36 (EQ-5D -0.0035 [95% CI -0.0153 to -0.0065], P_{Bayes} = 0.746; SF-36-MCS (-0.0111 [95% CI -0.0647 to 0.0416], P_{Bayes} = 0.70 and SF-36-PCS -0.0092 [95% CI -0.0758 to 0.0476], P_{Bayes} = 0.631.

Conclusions

Treatment effects on quality of life did not differ by multimorbidity (measured via a comorbidity count) at baseline—for the medical conditions studied, types and severity of comorbidities and level of quality of life at baseline, suggesting that evidence from clinical trials is likely to be applicable to settings with (at least modestly) higher levels of comorbidity.

Trial registration

A prespecified protocol was registered on PROSPERO (CRD42018048202).

Author summary

Why was this study done?

- The EuroQol-5 dimension (EQ-5D) and SF-36 are questionnaire-based tools that combine measures of physical and mental health into overall quality of life scores. These scores are used in randomised controlled trials of drug treatments to estimate how treatments affect quality of life. These estimates then inform decisions about which treatments should be offered to people with specific conditions.
- > Multimorbidity, the presence of 2 or more conditions, makes diagnosis and treatment more complex and is associated with worse quality of life in some settings.
- People with multimorbidity are underrepresented in clinical trials, and little is known about whether and how multimorbidity changes quality of life in clinical trials. This makes it difficult for clinicians and clinical guideline developers to determine how results from clinical trials should be applied to people with multimorbidity.

What did the researchers do and find?

- To address this uncertainty, we re-analysed data from existing clinical trials. Among 33,421 participants in 56 trials of new treatments for 16 different medical conditions, we used data on medication usage and medical histories to produce a comorbidity count (higher counts indicating more multimorbidity) for each individual.
- We then used statistical models to examine associations for comorbidity counts finding that higher comorbidity counts were associated with worse quality of life at trial entry and predicted a more rapid decline in quality of life over the course of the trial. However, having a higher comorbidity count did not change the effect of treatment on quality of life.

What do these findings mean?

- These findings suggest that where treatments improve quality of life for participants overall, they are similarly likely to do so for people with multimorbidity.
- Whether this is true for individuals with higher numbers of comorbidities or with conditions and treatments not included in our analyses remains uncertain.
- Nonetheless, our findings help inform clinical decision-making by reassuring clinicians, health economists, and guideline developers that overall trial results can be used when considering how best to manage many people with multimorbidity.

Introduction

Measures of health-related quality of life evaluate the impact of treatments in ways that matter to patients [1], and, by informing economic models, are important underpinnings of clinical guideline recommendations and therefore clinical practice. Generic quality of life measures, such as the EuroQol-5 dimension (EQ-5D) [2] and the 36-item short form survey (SF-36) [3] are valid across all conditions. This means they can be used to prioritise treatments even across different conditions. Consequently, quality of life measures are often included in randomised clinical trials (hereafter trials).

Multimorbidity, where individuals have 2 or more health conditions, is common and important as it is strongly associated with higher mortality and hospitalisation rates [4,5], and treatments that have been shown in clinical trials to improve quality of life are prescribed to people with multimorbidity less frequently than those without any comorbidities [6,7]. Cross-sectional associations between multimorbidity and quality of life have been reported [8–22]. Health-related quality of life has been identified as an essential core outcome in multimorbidity research [23]. Despite this, few studies have examined whether multimorbidity predicts change in quality of life over time [24–29], and whether the effect of treatments on quality of life differ according to the presence and extent of multimorbidity is unknown. Consequently, clinicians, regulatory agencies, and guideline developers are uncertain as to the applicability of trial quality of life findings for people with multimorbidity.

In previous individual participant data (IPD) analyses from a set of 124 clinical trials ranging across a number of index conditions and treatment comparisons, we have previously shown that multimorbidity, although underrepresented is present among trial participants and predicts increased rates of both serious adverse events and trial attrition [5,30,31]. Using the 56 trials from this set for which measures of quality of life are available, we now aim to determine whether comorbidity count, which is increased in multimorbidity, predicts change in quality life, and whether treatment effects on quality of life differ by comorbidity count at baseline.

Methods

This study is reported as per the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) guideline (<u>S1 PRISMA Checklist</u>).

Design

We performed a meta-analysis of trial IPD to determine the associations between comorbidity count and (i) quality of life at baseline; (ii) change in quality of life at trial follow-up; and (iii) the effect of treatment on change in quality of life at trial follow-up. Each analysis was done in 2 stages to account for the fact that trial data were stored securely on separate platforms and could not be analysed together in a single model. In the first stage, associations for individual trials were modelled within each secure trial analysis platform. In the second stage, the coefficients (and standard errors) from the first stage were meta-analysed using Bayesian linear models. We used a method that allowed partial pooling across index conditions and drug treatment comparisons in order to obtain overall drug treatment comparison specific and index-condition specific estimates of these associations.

Trial inclusion

Industry sponsored randomised trials with available IPD were identified using a prespecified protocol (Prospero CRD42018048202) [32]. The full selection process and analysis plan has been documented previously [30].

In brief, the United States clinical trial registry at clinicaltrials.gov [33] was searched for eligible studies meeting the following criteria; randomised phase 2/3, 3 or 4 trials studying preselected drug classes used to treat or prevent 23 selected long-term medical index conditions, registered from January 1990 until the dates of initial IPD access on 21 November 2016 and 18 May 2018, for Clinical Study Data Request (CSDR) and Yale University Open Data Access (YODA) repositories, respectively. Trials were excluded where they examined treatment for neoplastic, infectious, affective, psychotic, or developmental disorders. Trials were included if they were registered to clinicaltrials.gov [33] and had IPD availability within CSDR or YODA IPD repositories at the time of the original data request to these platforms. To make efficient use of analyst time (both ours and that of the trial sponsors who anonymised the IPD), included trials were limited to those with \geq 300 participants. Given our focus on comorbidity, trials were excluded if they limited participants to those aged less than 60 years old. For the current analysis, this set of trials was restricted to those with at least 1 generic health-related quality of life measure. As our aim in the current analysis was to examine the impact across a range of index conditions, we did not include trials which only included condition-specific measures of health-related quality of life (e.g., St. George's Respiratory Questionnaire). Included trials were categorised based on index medical condition and the World Health Organization Anatomic Therapeutic Chemical (ATC) drug class of the intervention drug [34].

Comorbidity data

A comorbidity count was calculated for each trial participant. Twenty-one conditions were included in the count: cardiovascular disease, chronic pain, arthritis, affective disorders, acid-

related disorders, asthma/chronic obstructive pulmonary disease, diabetes mellitus, osteoporosis, thyroid disease, thromboembolic disease, inflammatory conditions, benign prostatic hyperplasia, gout, glaucoma, urinary incontinence, erectile dysfunction, psychotic disorders, epilepsy, migraine, parkinsonism, and dementia.

The methods used to derive this count have been described previously [30,35]. Briefly, where only data on concomitant medication was available, specific ATC codes were used to define conditions (e.g., use inhaled of corticosteroids indicated the presence of asthma or chronic obstructive pulmonary disease). Where medical history data was also available, specific medical dictionary for regulatory activities (MedDRA) codes were additionally used to identify individuals with any of the 21 conditions. The comorbidity counts were summed for each participant. The full list of ATC and MedDRA codes and the analysis code used to derive the comorbidity count are available at the project's GitHub repository (https://github.com/ChronicDiseaseEpi/como_gol_public).

Outcome measures

Trials reported EQ-5D, SF-36, or both. EQ-5D is a preference-based generic health-related quality of life questionnaire with 5 domains (mobility, self-care, usual activities, pain/discomfort, and anxiety/depression) and a global measure of current health (measured via a visual analogue scale). Given EQ-5D is preference based, incorporating values for health outcomes, it can be used in cost-effectiveness analyses [36]. The domains are scored on either a 3 or 5 level scale (EQ-5D-3L or EQ-5D-5L). Since trials in our analysis included either type, but not both, we mapped each to a single EQ-5D index value (using United Kingdom population-based value sets for EQ-5D-3L and EQ-5D-5L, respectively, to obtain a single value that is comparable across all the EQ-5D trials [37]). EQ-5D index can range from negative values (reflecting a state felt to be "less preferable than death"), through 0 being "a state as bad as death" to 1 being "perfect health" [2].

The SF-36 questionnaire is also a commonly used generic quality of life instrument and includes 36 questions over 8 domains. Each domain score provides a weighted summary value of the questions within that domain. These domain scores are then summarised using a physical component score (PCS) and a mental component score (MCS). As is standard, we calculated these by first standardising the scales to z scores, then aggregating the physical and mental domains and transforming these into summary t scores ranging from 0 to 100 with higher scores denoting higher quality of life [3]. SF-36-PCS and SF-36-MCS scores are known to be correlated and therefore findings for each should not interpreted as if they provided independent verification [38].

Treatment arms

Treatment arm comparisons were specified prior to undertaking the quality of life analyses. For multi-arm trials, the most extreme arms were selected for comparison (e.g., if different dosages were used, the highest dose was compared to placebo or usual care—e.g., canagliflozin 300 mg, rather than 100 mg, versus placebo). Where placebo or usual care was included as a treatment arm, this was selected as the comparator. Otherwise, we chose the arm with the oldest treatment as the comparator arm.

Statistical analysis

Summary statistics were calculated for each index condition for the available EQ-5D and SF-36 trials including age (mean and standard deviation; SD), sex (number and %), and comorbidity count (mean and SD) and proportion with 0, 1 or 2, or more comorbidities. Full descriptions of the modelling are provided in the supplementary (<u>S1 Modelling</u> <u>description</u>) and are described briefly below.

(i) Association between comorbidity count and quality of life at baseline

In linear regression models, for each trial and each measure, baseline quality of life was modelled adjusting for age (per 15-year increment, which was close to the SD for most trials), sex (male versus referent group of females), and comorbidity count (per additional comorbidity). The effect measure estimates and associated standard errors for each model were then exported from the YODA and CSDR secure analysis platforms. In order to convert the measures onto a similar scale, we standardised each; we did so by dividing the estimates and standard errors by published estimates of the standard deviation (EQ-5D-index– 0.23 [39], SF-36-PCS—9.08, and SF-36-MCS– 10.16 [40]).

The effect measure estimates for comorbidity count (adjusted for age and sex) terms were then separately meta-analysed in Bayesian linear regression models. We used Bayesian models since these allowed partial pooling across index conditions and treatment comparisons and because they allowed us to obtain credible intervals for estimates at the level of index condition and treatment comparison directly from the posterior without a need for post hoc calculations. We fitted a range of meta-analyses, from the simplest where all trial-level estimates were pooled (ignoring treatment comparison and index condition), through models where there was partial pooling between either index conditions or treatment comparisons, to the most complex model where the estimates were partially pooled across both treatment comparisons and index condition. The Bayesian models were fitted using the *brms* package [41] in R statistical software. Samples from the posterior distribution for each association was summarised as the mean and the 2.5th and 97.5th percentiles (credible intervals) of the posterior distribution. The proportion of the distribution less than 0 (probability of negative association, i.e., Bayesian *P*) was also reported.

(ii) Association between comorbidity count and change in quality of life at trial follow-up

We obtained estimates similarly for the association between comorbidity count and change in health-related quality of life from baseline to trial follow-up by fitting the same models but with final quality of life score as the outcome and a term for the baseline score in the trial-level linear regression models.

(iii) Association between comorbidity count and the effect of treatment on change in quality of life at trial follow-up

Lastly, we obtained estimates for comorbidity count-treatment interactions for change in healthrelated quality of life from baseline to trial follow-up by fitting the same model as (ii) with additional terms for treatment arm and an arm-comorbidity count interaction. We repeated this model in a sensitivity analysis after excluding trials that did not demonstrate a benefit in quality of life.

In order to allow other researchers to use the treatment–covariate interaction results to inform subsequent analyses (e.g., as an informative prior), we obtained samples from the posterior. We obtained samples for index conditions and treatment comparisons included in our model, as well as for a notional new index condition and new treatment comparison not included in our model. We summarised these samples as student t-distributions. As with the main analysis, these models were fit using the *brms* package (<u>S1 Modelling description</u> for additional details).

Ethics approval and consent to participate

Ethical approval was obtained from the University of Glasgow, College of Medicine, Veterinary and Life Sciences ethics committee (200160070).

Results

Included trials

Of the 124 trials meeting our criteria [30,32], 56 provided 1 or more quality of life measures. Twenty-five trials (19,070 participants) provided EQ-5D only, 21 trials (8,595 participants) provided SF-36 only, and 10 trials provided both measures (5,756 participants) (Fig 1). One further trial that had collected EQ-5D and SF-36 was excluded as these results had been redacted by the study sponsors as part of the anonymisation process [42].

The characteristics of the 56 included trials were summarised by index condition (Table 1). The clinicaltrials.gov national clinical trial (NCT) identifiers of the included trials can be found in the Supporting information (Table A in <u>S1 Additional figures tables</u>). Index conditions were axial spondyloarthritis, chronic idiopathic urticaria, dementia, type 2 diabetes mellitus, pulmonary hypertension, inflammatory bowel disease, migraine, osteoporosis, Parkinson's disease, psoriasis, psoriatic arthropathy, pulmonary fibrosis, restless leg syndrome, rheumatoid arthritis, systemic lupus erythematosus (SLE), and thromboembolism. Not all arms were analysed from multi-arm trials; therefore, although included trials were limited to those with \geq 300 participants, the number of analysed participants per trial ranged from 102 to 2,568 participants. Approximately 82.1% of trials had a placebo or usual care arm as the comparator. The follow-up ranged from 6 to 104 weeks. Mean follow-up was 34.5 weeks. Thirty-seven trials and 19 trials were analysed on the CSDR and YODA platforms, respectively. The mean age of trial participants ranged from 38 years in SLE trials to 73 years in dementia trials and the percentage of male participants ranged from 6% in SLE trials to 79% in pulmonary fibrosis trials.

As we had access to IPD, we were able to make the following assessments of trial robustness to possible biases: 94.6% were at least double blinded (double blind n = 26, triple blind n = 8, quadruple blind n = 19), there being only 3 open label trials. Participant completion ranged from 38.4% to 100%, with a mean (SD) completion of 81.1% (13.2%) (median 83.3% and IQR 71.4% to 90.6%). Given this is an IPD analysis there was no reporting bias.

Comorbidity counts

As published previously, participants with comorbidities were present in trials for all index conditions [30]. Trials of selective immunosuppressants in SLE had the lowest proportion of participants with no comorbidities (6%) and the highest proportion of participants with ≥ 2 comorbidities (66%). Trials of tumour necrosis factor α (TNF- α) inhibitors in rheumatoid arthritis had the lowest proportion with ≥ 2 comorbidities (12%). Over half of participants in trials with SLE, pulmonary fibrosis, osteoporosis, pulmonary hypertension, and chronic idiopathic urticaria as the index condition, had 2 or more comorbidities.

Comorbidity count and quality of life at baseline

The mean baseline EQ-5D index scores and SF-36-PCS and SF-36-MCS scores varied across conditions and intervention drug classes (Table 1). On average, participants in trials of TNF- α drugs in rheumatoid arthritis had the lowest mean baseline EQ-5D index value at 0.74 (0.08). The highest mean baseline EQ-5D index value was found in participants from trials in type 2 diabetes mellitus of glucagon-like peptide-1 receptor agonists and sodium glucose co-transporter 2 inhibitors 0.92 (0.09).

The lowest SF-36 component scores were seen in SLE trials of selective immunosuppressants with a mean SF-36-PCS 27.9 (2.5) and SF-36-MCS 20.68 (2.09). The highest SF-36 component scores, SF-36-PCS 66.42 (13.37) and SF-36-MCS 63.67 (13.89), were seen in psoriasis trials of interleukin inhibitors.



Fig 1. PRISMA diagram for included trials in quality of life analyses. Flow diagram of included trials with IPD and reporting QoL measures; EQ-5D or the 36-item short form survey (SF-36). IPD, individual participant data; QoL, quality of life.

https://doi.org/10.1371/journal.pmed.1004154.g001

Across all index conditions and treatment comparisons, there was a negative association between comorbidity count and quality of life at baseline, when adjusted for age and sex, for all 3 measures: EQ-5D effect estimate -0.04 standardised units (95% CI -0.05, -0.02; Bayesian P > 0.999) per additional comorbidity; SF-36-PCS effect estimate -0.13 standardised units (95% CI -0.2, -0.06; Bayesian P > 0.999) per additional comorbidity; SF-36-MCS effect

Table 1. Participant characteristics by condition and drug class.

Trial index condition	QoL outcome	Drug treatment comparisons	n trials	Total participants	% Male	Mean age in years (SD)	Comorbidity count		ity	Mean baseline quality of life	
							0	1	≥2	EQ-5D index value (SD) or SF-36 PCS (SD); MCS (SD)	
Axial spondyloarthritis	EQ-5D	L04AC	1	102	75%	42 (12)	57%	24%	20%	0.76 (0.08)	
	SF-36	L04AB, L04AC	2	320	72%	40 (12)	49%	32%	19%	43.18 (11.46); 47.62 (13.95)	
Chronic idiopathic urticaria	EQ-5D	R03DX	3	653	28%	43 (14)	19%	27%	55%	0.89 (0.08)	
Dementia	EQ-5D	A10BG	3	2,265	41%	73 (8)	26%	30%	44%	0.88 (0.09)	
Diabetes mellitus, type 2	EQ-5D	A10BJ, A10BK	9	4,446	54%	57 (10)	31%	30%	40%	0.92 (0.09)	
	SF-36	A10BK	3	1,615	52%	56 (9)	26%	25%	49%	55.45 (9.01); 50.46 (7.64)	
Hypertension,	EQ-5D	G04BE	1	160	23%	53 (15)	22%	22%	55%	0.83 (0.09)	
pulmonary	SF-36	G04BE	1	160	23%	53 (15)	22%	22%	55%	44.98 (11.93); 54.87 (13.34)	
Inflammatory bowel	EQ-5D	L04AB	2	1,175	54%	40 (13)	43%	32%	24%	0.85 (0.08)	
disease	SF-36	L04AA, L04AB, L04AC	7	3,455	51%	39 (13)	42%	33%	25%	49.67 (11.46); 44.12 (12.97)	
Migraine	SF-36	N03AX	1	222	11%	40 (11)	-	-	-	63.48 (12.55); 59.9 (14.06)	
Osteoporosis	EQ-5D	H05AA, M05BA	4	4,377	45%	68 (13)	13%	26%	60%	0.85 (0.11)	
Parkinson's disease	EQ-5D	N04BC	1	343	55%	61 (10)	44%	32%	24%	0.78 (0.09)	
Psoriasis	EQ-5D	L04AC	3	1,117	68%	45 (13)	60%	24%	16%	0.83 (0.15)	
	SF-36	L04AC	2	960	68%	46 (12)	50%	29%	22%	66.42 (13.37); 63.67 (13.89)	
Psoriatic arthropathy	SF-36	L04AB, L04AC	3	597	55%	47 (11)	39%	35%	26%	43.67 (13.51); 49.58 (15.67)	
Pulmonary fibrosis	EQ-5D	L01XE	2	1,062	79%	67 (8)	19%	21%	60%	0.9 (0.1)	
Restless legs syndrome	SF-36	N04BC	1	331	40%	57 (12)	24%	34%	42%	63.55 (13.04); 60.79 (13.54)	
Rheumatoid arthritis	EQ-5D	L04AB	1	591	18%	52 (12)	61%	27%	12%	0.74 (0.08)	
	SF-36	L04AB, L04AC	9	4,608	20%	51 (12)	40%	31%	29%	42.31 (12.35); 47.92 (14.31)	
SLE	EQ-5D	L04AA	2	1,112	6%	38 (12)	6%	28%	66%	0.84 (0.1)	
	SF-36	L04AA	2	1,126	6%	38 (12)	6%	28%	66%	27.9 (2.5); 20.68 (2.09)	
Thromboembolism	EQ-5D	B01AE	3	6,450	59%	55 (16)	29%	29%	42%	0.85 (0.11)	

For the 1 included migraine trial, % of participants per comorbidity count are redacted due to small numbers in the interest of anonymity. See Fig 2 for ATC drug class. ATC, anatomic therapeutic chemical; MCS, mental component score; QoL, quality of life; SD, standard deviation; SLE, systemic lupus erythematosus.

https://doi.org/10.1371/journal.pmed.1004154.t001

SF-36-MCS

estimate -0.1 standardised units (95% CI -0.15, -0.04; Bayesian P = 0.999) per additional comorbidity.

Comorbidity count at baseline and subsequent change in quality of life

For all measures, relative to individuals without comorbidity, there was a decrease in quality of life over the course of the trial among participants with higher comorbidity counts. Table 2 shows these associations per one-unit increment in comorbidity count. The associations were similar in the simplest model where all trials were simply pooled and in the more complex models where trial was nested within index condition and/or drug class (Table 2). For EQ-5D, no model included the null (i.e., no association). For the SF-36 models, the results varied slightly by model. The 95% credible interval including the null for the model including drug class and index condition but did not include the null for the other models. However, even for the SF-36-PCS model, which had the widest 95% CI, the probability (Bayesian *P* value) that comorbidity count was negatively associated with quality of life (i.e., there was a smaller improvement/greater fall in quality of life in those trials where quality of life improved/worsened, respectively) was 95.6%.

There was no evidence of departure from linearity for the association between comorbidity count and quality of life for any of the measures (Table B in <u>S1 Additional figures tables</u>). This means that for a participant with a comorbidity count of 1 compared to a participant with a comorbidity count of 0 (or indeed for any given one-unit increment in comorbidity count), the increase in quality of life during the trial was 0.02 standardised units lower for EQ-5D (difference -0.02; 95% CI -0.03 to -0.01; Bayesian *P* value > 0.999), 0.05 units lower for SF-36-PCS, and 0.05 units lower for SF-36-MCS (<u>Table 2</u>). Back transforming these values to the original scales, this equates to 0.01 for the EQ-5D index, 0.48 for the SF-36-PCS, and 0.44 for

Model complexity	Model adjustment	EQ-5D index value	SF-36-PCS	
		35 trials, 24,826 participants	31 trials, 14,351 partici	
All trials pooled	Unadjusted	-0.03 (-0.03 to -0.02);	-0.06 (-0.11 to 0); P =	

Table 2. Cl	nange in quali	ty of life measures	from baseline to end	l of follow-up	by comorbidit	y count.
-------------	----------------	---------------------	----------------------	----------------	---------------	----------

		35 trials, 24,826 participants	31 trials, 14,351 participants	31 trials, 14,351 participants
All trials pooled	Unadjusted	-0.03 (-0.03 to -0.02); P > 0.999	-0.06 (-0.11 to 0); P = 0.978	-0.05 (-0.09 to 0); P = 0.97
	Age and sex adj.	-0.03 (-0.04 to -0.02); P > 0.999	-0.04 (-0.09 to 0.01); P = 0.962	-0.04 (-0.09 to 0.01); P = 0.965
Pooled by drug treatment comparisons	Unadjusted	-0.03 (-0.04 to -0.02); P > 0.999	-0.06 (-0.11 to 0); P = 0.972	-0.05 (-0.1 to 0); P = 0.975
	Age and sex adj.	-0.02 (-0.03 to -0.02); P > 0.999	$\begin{array}{c} -0.04 \ (-0.09 \ \text{to} \ 0.00); \\ P = 0.970 \end{array}$	-0.04 (-0.09 to 0.00); P = 0.967
Pooled by index condition	Unadjusted	-0.03 (-0.04 to -0.02); P > 0.999	-0.07 (-0.12 to -0.03); P = 0.996	-0.06 (-0.1 to -0.02); P = 0.993
	Age and sex adj.	-0.03 (-0.03 to -0.02); P > 0.999	-0.05 (-0.1 to -0.02); P = 0.995	-0.05 (-0.09 to -0.02); P = 0.997
Pooled by drug treatment comparisons and index condition	Unadjusted	-0.03 (-0.04 to -0.02); P > 0.999	-0.06 (-0.12 to 0); P = 0.971	-0.05 (-0.11 to 0.01); P = 0.965
	Age and sex adj.	-0.02 (-0.03 to -0.01); P > 0.999	$\begin{array}{c} -0.05 \ (-0.10 \ \text{to} \ 0.01); \\ P = 0.956 \end{array}$	-0.05 (-0.10 to 0.01); P = 0.966

Meta-analyses for the association between comorbidity count and change in quality of life scores from baseline to trial follow up. Where both EQ-5D and SF-36 were reported in a single trial, trial number and participants are included in both totals. Standardised effect estimates, 95% credibility intervals and Bayesian *P* values (probability of negative association).

MCS, mental component score; PCS, physical component score.

https://doi.org/10.1371/journal.pmed.1004154.t002



Fig 2. EQ-5D, SF-36-PCS, and SF-36-MCS: change in quality of life and treatment interaction by condition. The association of comorbidity and change in quality of life measures with treatment interaction (orange) and without (black). Standardised effect estimates with 50%, 80%, and 95% credibility intervals per trial index condition. EQ-5D, EuroQol-5 dimension; MCS, mental component score; PCS, physical component score; SF-36, the 36-item short form survey.

https://doi.org/10.1371/journal.pmed.1004154.g002

the SF-36-MCS. These associations between comorbidity count and change in quality of life are approximately similar in magnitude (per 1 additional comorbidity) to the treatment effects on change in quality of life for the 56 trials (Figure A in <u>S1 Additional figures tables</u>). Therefore, the presence of 2 or more comorbidities has a larger effect than treatment on quality of life. The associations were similar for different index conditions (<u>Fig 2</u>) and treatment comparisons (<u>Fig 3</u>).

Comorbidity count at baseline and variation in the effect of treatment on quality of life during trial follow-up

There was no evidence of an interaction between comorbidity count and treatment effect on EQ-5D, SF-36-PCS, or SF-36-MCS. The probability that treatment was less effective in improving quality of life in those with a greater number of comorbidities (Bayesian *P* values) ranged from 0.556 to 0.749 (<u>Table 3</u>). The point estimates for these interactions were an order of magnitude smaller than those for comorbidity count and change in quality of life (see



Fig 3. EQ-5D, SF-36-PCS and SF-36-MCS: change in quality of life and treatment interaction by treatment comparison. The association of comorbidity and change in quality of life measures with treatment interaction (orange) and without (black). Standardised effect estimates with 50%, 80%, and 95% credibility intervals per trial intervention ATC drug treatment comparisons. ATC, anatomic therapeutic chemical; EQ-5D, EuroQol-5 dimension; MCS, mental component score; PCS; physical component score; SF-36, the 36-item short form survey.

https://doi.org/10.1371/journal.pmed.1004154.g003

previous section). This was true across index conditions (Fig 2) and treatment comparisons (Fig 3), and the associations were null across all models from the simplest where all trials were pooled to the most complex where trial was nested within treatment comparison and index condition. Summarised posterior predictions (as student t-distributions) can be found in Table C in <u>S1 Additional figures tables</u>, for use by other researchers as informative priors for subsequent analyses. We also repeated the interaction analysis after excluding the trials (23 EQ-5D, 20 SF-36-PCS, and 23 SF-36-MCS) that did not demonstrate a benefit in quality of life; the findings were similarly null (Table D in <u>S1 Additional figures tables</u>).

Discussion

In an IPD meta-analysis of 56 clinical trials across 16 index conditions and 14 treatment comparisons, we found that a higher comorbidity count at baseline was associated with poorer quality of life at baseline (as measured by EQ-5D, SF-36-PCS, and SF-36-MCS). The higher

Model complexity	EQ-5D index value	SF-36-PCS	SF-36-MCS
	35 trials, 24,826 participants	31 trials, 14,351 participants	31 trials, 14,351 participants
All trials pooled	-0.0023 (-0.0091 to 0.0047)	-0.0042 (-0.0541 to 0.0409)	-0.0105 (-0.0508 to 0.0234)
	P = 0.747	P = 0.574	P = 0.749
Pooled by drug treatment comparisons	-0.0028 (-0.0117 to 0.0058)	-0.0043 (-0.0517 to 0.0375)	-0.0121 (-0.0536 to 0.0223)
	P = 0.746	P = 0.583	P = 0.763
Pooled by index condition	-0.0029 (-0.0126 to 0.0056) P = 0.743	-0.0027 (-0.0375 to 0.026) P = 0.556	-0.0115 (-0.0557 to 0.0288) P = 0.742
Pooled by drug treatment comparisons and index condition	-0.0035 (-0.0153 to 0.0065)	-0.0092 (-0.0758 to 0.0476)	-0.0111 (-0.0647 to 0.0416)
	P = 0.746	P = 0.631	P = 0.70

Table 3. Variation in the effect of treatment on quality of life by comorbidity count.

Meta-analyses for the interaction between comorbidity count and the intervention treatment effect on quality of life scores, where all models adjusted for age and sex. Where both EQ-5D and SF-36 were reported in a single trial, trial number and participants are included in both totals. Standardised effect estimates, 95% credibility intervals and Bayesian *P* values (probability of negative association).

MCS, mental component score; PCS, physical component score.

https://doi.org/10.1371/journal.pmed.1004154.t003

comorbidity count at baseline also predicted less improvement in EQ-5D over time. The findings were similar, but with wider 95% CIs which in some models just included the null, for SF-36-PCS and SF-36-MCS. However, baseline comorbidity count was not associated with differences in the estimated effect of treatment on quality of life for any of these measures.

Ours is the largest study, to our knowledge, to examine whether comorbidity predicts change in quality of life over time or in response to treatment in trial participants [24-29]. However, there are several limitations. First although comorbidity must, by definition increase with increases in multimorbidity, a simple count such as we used does not capture the full complexity of multimorbidity, including issues such as interactions between concordant and discordant conditions [43]. It is plausible that more nuanced measures of multimorbidity, feasible in trials designed to collect multimorbidity information prospectively, would lead to more nuanced results as to, for example, the effect of specific combinations of conditions on quality of life and heterogeneity of treatment effects. Moreover, while there was no evidence of any departure from linearity with the range of comorbidity counts we observed, this may not be true for higher comorbidity counts. Similarly, care should be taken when applying our findings to individuals with lower quality of life pretreatment, since the quality of life scores in the trial participants were generally fairly high at baseline. Secondly, the trials analysed were only those available via the CSDR and YODA trial repositories. Not all sponsors share data via these repositories, and among sponsors who do, not all trials are shared. Therefore, our dataset is not representative of all clinical trials, even for those index conditions and treatment comparisons included. As trials were identified from the US clinical trials register (clinicaltrials. gov) rather than a database of published papers (e.g., PubMed), there was no risk of publication bias. However, this does not mean that the trials were a random or complete sample of all registered trials. As we have noted previously [44], this set of trials were broadly similar to trials where IPD was not available with respect to the set indications, phases, number of participants, start dates, and exclusion criteria. However, inflammatory bowel disease and arthritis trials and trials of immunosuppressants were overrepresented, while phase 4 trials and those with especially large enrolment sizes were underrepresented.

Thirdly, while our definitions were prespecified, based on high-quality data, and have previously been used to demonstrate associations between comorbidity and both trial withdrawal and trial serious adverse events [5,31], the trials were not designed to measure comorbidity. It is possible that stronger associations may have been found if bespoke measures had been available. Thirdly, few of the trial participants had more than 2 comorbidities. As such, care should be taken in extrapolating our findings to individuals with 3 or more conditions as well as to people with more severe comorbidities who are likely to be excluded from trials, since it is possible that these types of comorbidities do modify treatment responses. Finally, our analysis was designed to examine variation in treatment effects by comorbidity count not overall treatment effects. In our modelling, we shared information (partial pooling) across multiple index conditions and treatment comparisons, pooling more extensively than would have been appropriate had our focus been on specific treatment comparisons. Consequently, our analysis of comorbidity treatment interactions should not be used to make inferences about the overall efficacy of specific treatment comparisons, but rather about plausible variation in treatment effects by comorbidity count.

A number of previous studies have examined cross-sectional associations between multimorbidity (or the presence of multiple comorbidities and quality of life, consistently showing negative associations [8–22]). Two systematic reviews with meta-analyses demonstrated a negative association; one looked at studies in any setting (mean reduction in quality of life for each additional comorbid condition -1.55% (95% CI -2.97% to -0.13%) and -4.37% (95% CI -7.13% to -1.61%) for mental and physical quality of life, respectively) [21], and one examined studies in non-hospital settings (reduction in quality of life 3.8% to 4.1% per additional comorbid condition) [22]. Multimorbidity has also been found to be associated with lower quality of life in the USA Medical Expenditure panel survey [19], a population-based survey of cancer patients in the Netherlands [20], a cross-sectional study of 1,649 people in primary care in India [8], and among 3,256 people with spinal osteoarthritis identified from Korean national health survey data [14].

In contrast, we found only a small number of studies that examined whether comorbidity or multimorbidity at baseline predicted longitudinal change in quality of life. In a cohort study of 1,211 adults in Japan, multimorbidity at baseline predicted more rapid decline in SF-36 over 12 months [27]. Similarly, among 1,582 people undergoing total hip arthroplasty in Denmark comorbidity at baseline was associated with a smaller increase in quality of life (high comorbidity versus no comorbidity EQ-5D change 0.09; 95% CI 0.02 to 0.16) [28]. In a study comprising 351 individuals attending an Australian clinic for people with complex chronic diseases, comorbidity count predicted more rapid decline in SF-36 (-0.11 per comorbidity)-0.96 to 0.76) [29]. However, all 3 studies had substantial loss to follow-up. In a secondary analysis of a clinical trial including 379 participants with early rheumatoid arthritis, quality of life measures were regressed on multimorbidity (measured via the rheumatoid arthritis comorbidity index) using linear mixed models. SF-36-PCS was found to decrease more quickly among participants with multimorbidity but there was no association for SF-36-MCS [25]. Two studies reported multimorbidity at baseline and reported longitudinal measures of quality of life in people with head and neck cancer and prostate cancer, respectively, but neither presented effect estimates or 95% confidence intervals making it difficult to judge whether this was due to the low sample sizes [24,26]. To this literature, we add the observation that in 33,421 participants in 56 trials across 16 index conditions, baseline multimorbidity is associated with lower quality of life at baseline and—at least for the EQ-5D-index—less improvement in quality of life associated with trial participation.

While we did not undertake a formal systematic review, in a wide-ranging search including a number of terms for comorbidity, multimorbidity, and quality of life (Table E in <u>S1 Addi-tional figures tables</u>), we found only 1 study that reported findings on variation of treatment effects on quality of life by comorbidity or multimorbidity. In a convenience sample of 3 trials (1 asthma, 1 heartburn, and 1 gastric ulcer disease) [45], the reported associations differed across the included trials, measures of quality of life and measures of multimorbidity. To this

sparse literature, we add findings, from a large number of participants, that there was no evidence of heterogeneity of treatment effect by multimorbidity for EQ-5D, SF-36-PCS, or SF-36-MCS.

The observation that treatment effects on quality of life did not differ by multimorbidity (measured via a comorbidity count) at baseline has implications for the interpretation of clinical trials. Unless there are strong a priori reasons to believe that an effect of a treatment on quality of life will be differential by comorbidity (e.g., a trial of a diabetes drug given by injection in a device that might be more difficult for people with arthritis to use may lead to differential efficacy in those people), reports of heterogeneity from individual trials and small meta-analyses should most likely be considered chance findings. Similarly, treatments that involve complex regimes may also differentially impact quality of life according to the participant's comorbidity count. For example, individuals with more comorbidities (and hence greater polypharmacy) may find additional treatments either more or less burdensome than individuals with fewer comorbidities. Our findings imply that treatment efficacy estimates for EQ-5D, SF-36-PCS, and SF-36-MCS from clinical trials are also likely to be applicable to settings with (at least modestly) higher levels of comorbidity, at least for the kinds of conditions and drug classes covered in our analysis.

Based on our models, for each outcome, we produced posterior predictions for comorbidity-treatment interactions for an unobserved drug class and index condition (i.e., one not included in our models). These distributions represent an estimate of the likely variation in treatment effect according to comorbidity count before seeing the trial data for any given specific treatment comparison. We had originally intended to repeat the modelling after logtransforming quality of life measures. However, since there were no interactions on the linear scale (which is standard for these measures), we did not examine the effect of changing scale. As such, these distributions can be used as informative priors for subsequent meta-analyses examining treatment effects in people with comorbidity, improving the reliability and precision of the resultant estimates. The prior distributions we produced can also be used to inform choices in probabilistic health economic models that are used to apply trial findings to "realworld" populations, such as in Health Technology Assessments [46]. For example, if a trial was conducted among participants with fewer comorbidities than patients in the target population, our findings can be used to model the likely effect of such treatments on quality of life in realworld settings. We observed a decline in quality of life in participants with comorbidities, relative to other participants. This finding provides new evidence to suggest that the association between comorbidity and quality of life is causal. Most previous studies reporting associations between multimorbidity or comorbidity and quality of life have been cross-sectional $[\underline{8}-\underline{22}]$, and the few longitudinal studies have been limited by small sizes [24-26], limited numbers of baseline conditions, or substantial loss to follow-up [27-29]. Our findings therefore strengthen the evidence for causation between comorbidity and baseline and subsequent change in quality of life.

There are a number of possible mechanisms for the relative decline in quality of life among people with multiple comorbidities. First, the decline may be a direct result of the underlying conditions (alone or in combination); symptoms such as pain, impaired sleep, limited mobility, and impaired function may increasingly adversely impact quality of life over time. Alternatively, the (non-treatment related) benefits of trial participation such as improved access to clinical care may be different in people with and without comorbidity. Similarly, attending visits, undergoing procedures, and following treatment regimens may also impose a greater treatment burden among people with comorbidity. This concept of trial participation burden is analogous to the treatment burden (visits, drug regimens, etc.) described in routine clinical practice [47], which is known to be more challenging for people with multimorbidity. In these clinical trials, higher comorbidity count is associated with lower quality of life at baseline and predicts subsequent relative decline in EQ-5D (and most likely SF-36) over time. However, the effect of treatment on quality of life does not differ by comorbidity count at baseline. Trial-derived estimates for treatment effects for quality of life are likely to be applicable to people with moderate numbers of comorbidities.

Supporting information

S1 PRISMA checklist. PRISMA checklist. This file is a PRISMA checklist for the manuscript. (DOCX)

S1 Modelling description. Detailed description of modelling. This file includes a further detailed description of the statistical modelling performed. (DOCX)

S1 Additional figures tables. Supplementary figures and tables. This file contains supplementary figures and tables. (DOCX)

Acknowledgments

This study, carried out under YODA Project # 2017–1746, used data obtained from the Yale University Open Data Access Project, which has an agreement with JANSSEN RESEARCH & DEVELOPMENT, L.L.C. The interpretation and reporting of research using this data are solely the responsibility of the authors and does not necessarily represent the official views of the Yale University Open Data Access Project or JANSSEN RESEARCH & DEVELOPMENT, L.L.C. This study was also carried out under <u>ClinicalStudyDataRequest.com</u> project number 1732, used data from the <u>ClinicalStudyDataRequest.com</u> repository, who provided data from Boehringer-Ingelheim, GSK, Lilly, Roche, Takeda, and Sanofi. The interpretation and reporting of research using these data are solely the responsibility of the authors and does not necessarily represent the official views of <u>ClinicalStudyDataRequest.com</u> or Boehringer-Ingelheim, GSK, Lilly, Roche, Takeda, or Sanofi.

Author Contributions

Conceptualization: Peter Hanlon, Anoop S. V. Shah, Emma McIntosh, Jim Lewsey, Sarah H. Wild, Bruce Guthrie, David M. Kent, Sofia Dias, Nicky J. Welton, David A. McAllister.

Data curation: Elaine W. Butterly, Laurie J. Hannigan, David A. McAllister.

Formal analysis: Elaine W. Butterly, David A. McAllister.

Funding acquisition: David A. McAllister.

Investigation: David A. McAllister.

Methodology: Emma McIntosh, Sarah H. Wild, Frances S. Mair, Sofia Dias, Nicky J. Welton, David A. McAllister.

Project administration: Elaine W. Butterly, David A. McAllister.

Resources: David A. McAllister.

Software: David A. McAllister.

Supervision: David A. McAllister.

Validation: David A. McAllister.

Visualization: David A. McAllister.

Writing - original draft: Elaine W. Butterly.

Writing – review & editing: Elaine W. Butterly, Peter Hanlon, Anoop S. V. Shah, Laurie J. Hannigan, Emma McIntosh, Jim Lewsey, Sarah H. Wild, Bruce Guthrie, Frances S. Mair, David M. Kent, Sofia Dias, Nicky J. Welton, David A. McAllister.

References

- Sherbourne CD, Sturm R, Wells KB. What Outcomes Matter to Patients?. J Gen Intern Med. 1999; 14:357–363. <u>https://doi.org/10.1046/j.1525-1497.1999.00354.x</u> PMID: <u>10354256</u>
- EQ-5D User Guides–EQ-5D. [cited 2021 Nov 19]. Available from: <u>https://euroqol.org/publications/user-guides/</u>.
- 3. Ware JEJ. SF-36 Health Survey Update. Spine. 2000; 25:3130–3139. <u>https://doi.org/10.1097/00007632-200012150-00008</u> PMID: <u>11124729</u>
- Wen Y-C, Chen L-K, Hsiao F-Y. Predicting mortality and hospitalization of older adults by the multimorbidity frailty index. PLoS ONE. 2017; 12:e0187825. <u>https://doi.org/10.1371/journal.pone.0187825</u> PMID: <u>29145407</u>
- Hanlon P, Butterly E, Shah AS, Hannigan LJ, Wild SH, Guthrie B, et al. Assessing trial representativeness using Serious Adverse Events: An observational analysis using aggregate and individual-level data from clinical trials and routine healthcare data. BMC Med. 2022. <u>https://doi.org/10.1186/s12916-022-02594-9</u> PMID: 36303169
- Smith DJ, Martin D, McLean G, Langan J, Guthrie B, Mercer SW. Multimorbidity in bipolar disorder and undertreatment of cardiovascular disease: a cross sectional study. BMC Med. 2013; 11:263. <u>https://doi.org/10.1186/1741-7015-11-263 PMID: 24359325</u>
- Wang PS, Avorn J, Brookhart MA, Mogun H, Schneeweiss S, Fischer MA, et al. Effects of noncardiovascular comorbidities on antihypertensive use in elderly hypertensives. Hypertens Dallas Tex. 1979; 2005(46):273–279. <u>https://doi.org/10.1161/01.HYP.0000172753.96583.e1</u>
- Pati S, Swain S, Knottnerus JA, Metsemakers JFM, van den Akker M. Health related quality of life in multimorbidity: a primary-care based study from Odisha. India Health Qual Life Outcomes. 2019; 17:116. https://doi.org/10.1186/s12955-019-1180-3 PMID: 31277648
- Fortin M, Dubois M-F, Hudon C, Soubhi H, Almirall J. Multimorbidity and quality of life: a closer look. Health Qual Life Outcomes. 2007; 5:52. <u>https://doi.org/10.1186/1477-7525-5-52</u> PMID: <u>17683600</u>
- Fortin M, Lapointe L, Hudon C, Vanasse A, Ntetu AL, Maltais D. Multimorbidity and quality of life in primary care: a systematic review. Health Qual Life Outcomes. 2004; 2:51. <u>https://doi.org/10.1186/1477-7525-2-51</u> PMID: <u>15380021</u>
- 11. Brettschneider C, Leicht H, Bickel H, Dahlhaus A, Fuchs A, Gensichen J, et al. Relative Impact of Multimorbid Chronic Conditions on Health-Related Quality of Life–Results from the MultiCare Cohort Study. PLoS ONE. 2013; 8:e66742. https://doi.org/10.1371/journal.pone.0066742 PMID: 23826124
- Chen W, Lynd LD, FitzGerald JM, Marra CA, Rousseau R, Sadatsafavi M. The added effect of comorbidity on health-related quality of life in patients with asthma. Qual Life Res Int J Qual Life Asp Treat Care Rehabil. 2015; 24:2507–2517. <u>https://doi.org/10.1007/s11136-015-0995-6</u> PMID: <u>26038225</u>
- Fraser SD, Barker J, Roderick PJ, Yuen HM, Shardlow A, Morris JE, et al. Health-related quality of life, functional impairment and comorbidity in people with mild-to-moderate chronic kidney disease: a crosssectional study. BMJ Open. 2020; 10:e040286. <u>https://doi.org/10.1136/bmjopen-2020-040286</u> PMID: <u>32764092</u>
- 14. Kim S-K, Choe J-Y. Comorbidities and Health-Related Quality of Life in Subjects with Spine Osteoarthritis at 50 Years of Age or Older: Data from the Korea National Health and Nutrition Examination Survey. Medicina (Kaunas). 2022; 58:126. <u>https://doi.org/10.3390/medicina58010126</u> PMID: <u>35056434</u>
- 15. Adams RJ, Wilson DH, Taylor AW, Daly A, Tursan d'Espaignet E, Dal Grande E, et al. Coexistent chronic conditions and asthma quality of life: a population-based study. Chest. 2006; 129:285–291. https://doi.org/10.1378/chest.129.2.285 PMID: 16478843
- Al-Asadi AM, Klein B, Meyer D. Multiple comorbidities of 21 psychological disorders and relationships with psychosocial variables: a study of the online assessment and diagnostic system within a webbased population. J Med Internet Res. 2015; 17:e55. <u>https://doi.org/10.2196/jmir.4143</u> PMID: 25803420

- Millá-Perseguer M, Guadalajara-Olmeda N, Vivas-Consuelo D, Usó-Talamantes R. Measurement of health-related quality by multimorbidity groups in primary health care. Health Qual Life Outcomes. 2019; 17:8. https://doi.org/10.1186/s12955-018-1063-z PMID: 30634992
- Nelis SM, Wu Y-T, Matthews FE, Martyr A, Quinn C, Rippon I, et al. The impact of co-morbidity on the quality of life of people with dementia: findings from the IDEAL study. Age Ageing. 2019; 48:361–367. https://doi.org/10.1093/ageing/afy155 PMID: 30403771
- Sullivan PW, Ghushchyan VH, Bayliss EA. The impact of co-morbidity burden on preference-based health-related quality of life in the United States. Pharmacoeconomics. 2012; 30:431–442. <u>https://doi.org/10.2165/11586840-00000000-00000</u> PMID: 22452633
- Vissers PAJ, Thong MSY, Pouwer F, Zanders MMJ, Coebergh JWW, van de Poll-Franse LV. The impact of comorbidity on Health-Related Quality of Life among cancer survivors: analyses of data from the PROFILES registry. J Cancer Surviv Res Pract. 2013; 7:602–613. <u>https://doi.org/10.1007/s11764-013-0299-1</u> PMID: 23918453
- Makovski TT, Schmitz S, Zeegers MP, Stranges S, van den Akker M. Multimorbidity and quality of life: Systematic literature review and meta-analysis. Ageing Res Rev. 2019; 53:100903. <u>https://doi.org/10.1016/j.arr.2019.04.005</u> PMID: <u>31048032</u>
- Schmitz S, Makovski TT, Adams R, van den Akker M, Stranges S, Zeegers MP. Bayesian Hierarchical Models for Meta-Analysis of Quality-of-Life Outcomes: An Application in Multimorbidity. Pharmacoeconomics. 2020; 38:85–95. <u>https://doi.org/10.1007/s40273-019-00843-z</u> PMID: <u>31583600</u>
- Smith SM, Wallace E, Salisbury C, Sasseville M, Bayliss E, Fortin M. A Core Outcome Set for Multimorbidity Research (COSmm). Ann Fam Med. 2018; 16:132–138. <u>https://doi.org/10.1370/afm.2178</u> PMID: 29531104
- Daskivich TJ, van de Poll-Franse LV, Kwan L, Sadetsky N, Stein DM, Litwin MS. From bad to worse: comorbidity severity and quality of life after treatment for early-stage prostate cancer. Prostate Cancer Prostatic Dis. 2010; 13:320–327. <u>https://doi.org/10.1038/pcan.2010.33</u> PMID: <u>20838413</u>
- Stouten V, Westhovens R, De Cock D, Van der Elst K, Pazmino S, Bertrand D, et al. Having a co-morbidity predicts worse outcome in early rheumatoid arthritis despite intensive treatment: a post hoc evaluation of the pragmatic randomized controlled CareRA trial. Rheumatology (Oxford). 2021; 60:3699– 3708. <u>https://doi.org/10.1093/rheumatology/keaa841</u> PMID: <u>33434277</u>
- Oozeer NB, Benbow J, Downs C, Kelly C, Welch A, Paleri V. The effect of comorbidity on quality of life during radiotherapy in head and neck cancer. Otolaryngol Head Neck Surg. 2008; 139:268–272. <u>https:// doi.org/10.1016/j.otohns.2008.05.027</u> PMID: <u>18656727</u>
- Aoki T, Fukuhara S, Fujinuma Y, Yamamoto Y. Effect of multimorbidity patterns on the decline in health-related quality of life: a nationwide prospective cohort study in Japan. BMJ Open. 2021; 11: e047812. <u>https://doi.org/10.1136/bmjopen-2020-047812</u> PMID: <u>34127493</u>
- Glassou EN, Pedersen AB, Aalund PK, Mosegaard SB, Hansen TB. Is gain in health-related quality of life after a total hip arthroplasty depended on the comorbidity burden? Acta Orthop. 2018; 89:374–379. https://doi.org/10.1080/17453674.2018.1457885 PMID: 29621916
- Tyack Z, Frakes K, Barnett A, Cornwell P, Kuys S, McPhail S. Predictors of health-related quality of life in people with a complex chronic disease including multimorbidity: a longitudinal cohort study. Qual Life Res. 2016; 25:2579–2592. https://doi.org/10.1007/s11136-016-1282-x PMID: 27048497
- Hanlon P, Hannigan L, Rodriguez-Perez J, Fischbacher C, Welton NJ, Dias S, et al. Representation of people with comorbidity and multimorbidity in clinical trials of novel drug therapies: an individual-level participant data analysis. BMC Med. 2019; 17:201. <u>https://doi.org/10.1186/s12916-019-1427-1</u> PMID: <u>31711480</u>
- Lees JS, Hanlon P, Butterly EW, Wild SH, Mair FS, Taylor RS, et al. Effect of age, sex, and morbidity count on trial attrition: meta-analysis of individual participant level data from phase 3/4 industry funded clinical trials. BMJ Med. 2022:1. https://doi.org/10.1136/bmjmed-2022-000217
- Prospero registration. [cited 2022 Feb 22]. Available from: <u>https://www.crd.york.ac.uk/PROSPERO/</u> display_record.php?ID=CRD42018048202&ID=CRD42018048202.
- 33. Home—ClinicalTrials.gov. [cited 2020 Jul 1]. Available from: https://clinicaltrials.gov/.
- 34. WHOCC—ATC/DDD Index. [cited 2022 Feb 22]. Available from: https://www.whocc.no/atc_ddd_index/
- Hanlon P, Jani BD, Butterly E, Nicholl B, Lewsey J, McAllister DA, et al. An analysis of frailty and multimorbidity in 20,566 UK Biobank participants with type 2 diabetes. Commun Med. 2021; 1:1–9. <u>https:// doi.org/10.1038/s43856-021-00029-9</u> PMID: <u>35602215</u>
- Neumann PJ, Goldie SJ, Weinstein MC. Preference-based measures in economic evaluation in health care. Annu Rev Public Health. 2000; 21:587–611. <u>https://doi.org/10.1146/annurev.publhealth.21.1.587</u> PMID: <u>10884966</u>

- Valuation of EQ-5D –EQ-5D. [cited 2022 Oct 14]. Available from: <u>https://euroqol.org/eq-5d-instruments/valuation-of-eq-5d/</u>.
- Taft C, Karlsson J, Sullivan M. Do SF-36 Summary Component Scores Accurately Summarize Subscale Scores? Qual Life Res. 2001; 10:395–404. <u>https://doi.org/10.1023/a:1012552211996</u> PMID: <u>11763202</u>
- Kind P, Hardman G, Macran S. UK population norms for EQ-5D. Centre for Health Economics, University of York; 1999 Nov. Report No.: 172chedp. Available from: <u>https://ideas.repec.org/p/chy/respap/172chedp.html</u>.
- 40. Garratt AM, Stavem K. Measurement properties and normative data for the Norwegian SF-36: results from a general population survey. Health Qual Life Outcomes. 2017; 15:51. <u>https://doi.org/10.1186/ s12955-017-0625-9</u> PMID: <u>28292292</u>
- 41. brms: An R Package for Bayesian Multilevel Models Using Stan. J Stat Softw. [cited 2022 Feb 22]. Available from: https://www.jstatsoft.org/article/view/v080i01.
- Anonymisation of Clinical Trial Datasets. Available from: <u>https://clinicalstudydatarequest.com/</u> Documents/Anonymisation%200f%20Clinical%20Trial%20Datasets.pdf.
- **43.** Overview | Multimorbidity: clinical assessment and management | Guidance | NICE. [cited 2022 Oct 14]. Available from: https://www.nice.org.uk/guidance/ng56.
- 44. Hanlon P, Hannigan L, Rodriguez-Perez J, Fischbacher C, Welton NJ, Dias S, et al. Representation of people with comorbidity and multimorbidity in clinical trials of novel drug therapies: an individual-level participant data analysis. BMC Med. 2019; 17:201– Appendix 8. Available from: <u>https://static-content.springer.com/esm/art%3A10.1186%2Fs12916-019-1427-1/MediaObjects/12916_2019_1427_MOESM8_ESM.pdf. https://doi.org/10.1186/s12916-019-1427-1 PMID: 31711480</u>
- **45.** Xuan J, Kirchdoerfer LJ, Boyer JG, Norwood GJ. Effects of comorbidity on health-related quality-of-life scores: an analysis of clinical trial data. Clin Ther. 1999; 21:383–403. <u>https://doi.org/10.1016/S0149-2918(00)88295-8 PMID: 10211540</u>
- 46. Dias S, Welton NJ, Sutton AJ, Ades AE. Evidence Synthesis for Decision Making 1: Introduction. Med Decis Making. 2013; 33:597–606. <u>https://doi.org/10.1177/0272989X13487604</u> PMID: <u>23804506</u>
- 47. Mair FS, May CR. Thinking about the burden of treatment. BMJ. 2014; 349:g6680. <u>https://doi.org/10.1136/bmj.g6680</u> PMID: 25385748